

GOPEN ACCESS

Citation: Abeijon C, Alves F, Monnerat S, Mbui J, Viana AG, Almeida RM, et al. (2020) Urine-based antigen detection assay for diagnosis of visceral leishmaniasis using monoclonal antibodies specific for six protein biomarkers of *Leishmania infantum* / *Leishmania donovani*. PLoS Negl Trop Dis 14(4): e0008246. https://doi.org/10.1371/journal. pntd.0008246

Editor: Camila I. de Oliveira, Instituto Goncalo Moniz-FIOCRUZ, BRAZIL

Received: November 7, 2019

Accepted: March 24, 2020

Published: April 23, 2020

Copyright: © 2020 Abeijon et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: This work was supported by a grant from the National Institutes of Health to Claudia Abeijon (R44-Al113992). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

RESEARCH ARTICLE

Urine-based antigen detection assay for diagnosis of visceral leishmaniasis using monoclonal antibodies specific for six protein biomarkers of *Leishmania infantum / Leishmania donovani*

Claudia Abeijon¹, Fabiana Alves², Séverine Monnerat², Jane Mbui³, Agostinho G. Viana³, Raquel M. Almeida⁴, Lilian L. Bueno⁴, Ricardo T. Fujiwara⁴, Antonio Campos-Neto^{1,5}*

1 DetectoGen Inc., Grafton, Massachusetts, United States of America, 2 Drugs for Neglected Diseases *Initiative*, Geneva, Switzerland, 3 KEMRI Centre for Clinical Research, Nairobi, Kenya, 4 Federal University of Minas Gerais, Belo Horizonte, MG, Brazil, 5 Cummings School of Veterinary Medicine at Tufts University, North Grafton, Massachusetts, United States of America

* acampos@detectogen.com, Antonio.Campos@tufts.edu

Abstract

The development of an accurate protein-based antigen detection assay for diagnosis of active visceral leishmaniasis (VL) would represent a major clinical advance. VL is a serious and fatal disease caused by the parasites Leishmania infantum and Leishmania donovani. The gold standard confirmatory diagnostic test for VL is the demonstration of parasites or their DNA from aspirates from spleen, lymph node, and bone marrow or from blood buffy coats. Here we describe the production and use of monoclonal antibodies (mAbs) for the development of a sensitive and specific antigen detection capture ELISA for VL diagnosis. This test simultaneously detects six leishmania protein biomarkers that we have previously described (Li-isd1, Li-txn1, Li-ntf2, Ld-mao1, Ld-ppi1 and Ld-mad1). The initial clinical validation of this new mAb-based multiplexed capture ELISA showed a sensitivity of \geq 93%. The test was negative with 35 urine samples from healthy control subjects as well as with 30 patients with confirmed non-VL tropical diseases (cutaneous leishmaniasis, n = 6; Chagas disease, n = 6; schistosomiasis, n = 6; and tuberculosis, n = 12). These results strongly support the possible utility of this mAb-based multiplexed capture ELISA as a promising diagnostic test for active VL as well as for monitoring the treatment efficacy of this disease. The test is ready for upscaling and validation for clinical use.

Author summary

Visceral leishmaniasis (VL) or kala-azar is a systemic parasitic disease that is endemic in 75 countries with more than 500 million people at risk of infection. VL is caused primarily by the parasites known as *Leishmania infantum* and *Leishmania donovani*. It is estimated that VL affects 50,000–90,000 people each year, with 90% of the cases occurring in India,

Competing interests: CA is an employee of DetectoGen Inc. and has an ownership option in the company. AC-N is a co-founder/director of DetectoGen Inc. and has ownership of the company. None of the other co-authors has any financial conflict of interest.

Kenya, Somalia, Sudan, South Sudan and Brazil. VL is a fatal disease if not treated. Laboratory diagnosis of this disease can be performed using several different approaches. However, there is still a need for a simple, non-expensive, sensitive and specific test that can be used not only for the accurate diagnosis of this serious disease but also to monitor the efficacy of its treatment. Here we describe a new test that can circumvent most of the drawbacks of existing approaches. This new test detects six leishmanial proteins or biomarkers that are eliminated in the urine of patients with VL. The test is a conventional ELISA assembled with monoclonal antibodies (mAbs) specific for each of the six different biomarkers and it is performed in urine samples of suspected patients. An initial clinical validation demonstrated that the test has a sensitivity of \geq 93% and specificity of 100%. This new and accurate mAb-based multiplexed assay could be a useful resource to diagnose most clinical forms and/or the severity of the disease e.g., active VL, asymptomatic VL, post kala-azar dermal leishmaniasis, VL/HIV co-infection, etc.

Introduction

Visceral leishmaniasis (VL) is a serious parasitic disease caused by *Leishmania donovani* / *Leishmania infantum* [1, 2]. Diagnosis of VL is routinely performed by invasive splenic, bone marrow, or lymph node aspiration, followed by microscopic identification of the parasites and or culture. The sensitivity of these tests is in general modest and varies enormously. Nucleic acid amplification tests have better sensitivity [3] but are relatively more complicated, expensive, and are restricted to referral hospitals and advanced research centers [4–12]. Due to the limitations of these invasive approaches, the presence of anti-parasite antibodies in serum is also routinely used to diagnose VL [13, 14]. However, antibody tests have variable sensitivity in different endemic regions [15–17], and cannot discriminate active disease from cured individuals. An antigen detection test that detects parasite carbohydrate antigens in urine of VL patients with active disease was developed several years ago [18–21]. Unfortunately, the sensitivity/specificity of the test varied widely, probably due to the heterogeneity of the parasites' carbohydrate antigens.

We have recently developed an alternative approach to circumvent these restrictions: a multiplexed capture ELISA that detects the *L. donovani / L. infantum* protein biomarkers *Liisd1, Li-txn1, Li-ntf2, Ld-mao1* and *Ld-ppi1* [22]. These proteins were previously discovered using mass spectroscopy in the urine of VL patients [23–25]. The multiplexed assay was assembled with polyclonal rabbit IgG and chicken IgY antibodies specific for these five antigens and showed a sensitivity of 82.2% for the diagnosis of VL. A sixth biomarker (*Ld-mad1*), which was also initially discovered, was difficult to express as a recombinant protein and was not evaluated in these former studies. Although polyclonal antibody-based tests (in contrast to monoclonal antibodies) are not ideal for upscaling and commercial production, these proof-of concept and preliminary results validated the utility of the discovered leishmanial protein biomarkers found in urine of VL patients as powerful tools for the development of an accurate diagnostic test for this disease.

Here we report the development of a monoclonal antibody-based multiplexed capture ELISA that detects the six VL protein biomarkers previously described. The initial clinical validation of this new VL multiplexed test showed a sensitivity of \geq 93% and a specificity of 100%.

Material and methods

Clinical specimens

A total of 24 urine samples of VL patients from Brazil, 12 females and 12 males, ages 2-65y, assumed to be infected with *L. infantum*, and 45 urine samples of VL patients from Kenya, 18 females and 27 males, ages 6-36y assumed to be infected with *L. donovani*, were evaluated in this study. These samples were collected before the initiation of therapy and were from patients diagnosed with VL based on the following criteria: a clinical course consistent with VL (e.g., fever, anemia, hepatosplenomegaly), and confirmatory laboratory findings (identification of *Leishmania* in spleen or bone marrow aspirates) and positive serological test. None of the patients had any clinical symptoms or laboratory findings compatible with renal or urinary tract abnormalities, nor were any of them receiving anti *Leishmania* therapy at the time of urine collection. In addition, none of the VL patiens were positive for HIV.

Ethics statement

All samples from Brazil (VL patients and controls) were obtained from the University Hospital Clemente Farias (Montes Claros, Minas Gerais, Brazil). Clearance approval to use these samples was obtained from the Human Research Ethics Committee—COEP (CAAE -00842112.2.0000.5149) of the Federal University of Minas Gerais. The samples from Kenya were obtained from Kacheliba County Hospital (West Pokot County) and from Kimalel Health Center (Baringo County). Clearance approval to use these samples was obtained from the KEMRI Scientific and Ethics Review Unit (KEMRI/SERU/CCR/0011/3120). The control samples included 35 urine samples obtained from healthy control subjects living in the same geographical areas as the VL patients. In addition, control samples from non-VL patients from Brazil who had other infectious diseases (cutaneous leishmaniasis, n = 6; Chagas disease, n = 6; schistosomiasis, n = 6; and tuberculosis, n = 12) were also included. The serological tests for VL were negative in all control samples. All samples used in this study were anonymized. The overall data analysis plan for the study is illustrated in Fig 1.

Leishmania donovani protein expression and purification

The DNA sequence coding for the *L. donovani Ld-mad1* was codon optimized for expression in *E. coli*. The gene was synthesized by Blue Heron (Bothell, WA). To allow sub-cloning, restriction enzyme sequences Nde I and Bam HI were included at the 5' and 3' endings, respectively, of the optimized DNA fragment. The synthetic gene was digested with the restriction enzymes and sub-cloned into a pET-14b expression vector, which was similarly digested for directional cloning. Protein expressed in the pET-14b expression vector contains a six-residue histidine tag at the N-terminus of the molecule, which facilitates purification by affinity on QIAexpress Ni-NTA agarose matrix (Qiagen, Valencia, CA). Recombinant protein was obtained and purified from 100ml of IPTG-induced batch cultures by affinity chromatography using QIAexpress Ni-NTA agarose matrix (QIAGEN, Chatsworth, CA) as described [26]. The yields of recombinant proteins were 10-20mg per liter of induced bacterial culture, and purity was assessed by SDS-PAGE followed by Coomassie blue staining.

Generation of specific monoclonal antibodies (mAbs)

The generation and purification of the mAbs were outsourced to GenScript USA Inc, Piscataway, NJ 08854 USA and followed standard procedures [27, 28]. Briefly, the purified recombinant proteins *Li-isd1*, *Li-txn1*, *Li-ntf2*, *Ld-mao1*, *Ld-ppi1* and *Ld-mad1* (50µg of each) were individually emulsified with an equal volume of complete Freund's adjuvant and injected



Fig 1. Diagram of data analysis plan.

https://doi.org/10.1371/journal.pntd.0008246.g001

subcutaneously into three C57BL/6 mice per antigen. The animals were given two subcutaneous boosters (25µg of protein in IFA) two weeks apart. One week after the first boost the animals were bled and serum was collected and tested by ELISA to determine the titer of each antiserum. The mouse producing the highest titer of IgG specific for each marker was selected for production of the mAbs. The mice were sacrificed three days after the second boost, their spleens were harvested and the spleen cells were fused with the mieloma cell line SP2/0 for generation of hybridomas. Monoclonal hybridoma clones were then obtained by limiting dilution and their supernatants were tested for the presence of specific IgG antibody using both quantitative ELISA and Western blot analysis. Twenty clones were selected for each marker. IgG mAbs were purified from the hybridoma supernatants by affinity chromatography using

recombinant protein A/G immobilized resin [29]. Aliquots of selected IgG mAbs were subsequently biotin labeled [30].

Epitope recognition by mAbs

Supernatants from each of the 20 hybridoma clones were tested for their reactivity with synthetic purified 20mer peptides covering the entire full length of each biomarker and overlapping by 10 amino acids. Reactivity was tested by direct ELISA as previously described [31]. Peptides were synthesized by GenScript (Piscataway, NJ).

Western Blot

Purified recombinant Ld-mad1 (50 ng) and whole lysate extract from L. donovani amastigotes and promastigotes were fractionated by SDS-PAGE (4-20% gradient gel) and transferred to polyvinylidene fluoride membrane (PVDF, Millipore, Medford, MA). Whole lysate of L. donovani from promastigote parasites was prepared from the organisms cultured for 7-10 days in complete Schneider's medium at 26°C. Whole lysate of L. donovani amastigotes was from in vitro infected macrophage cell line DH82 [32]. Briefly, macrophages were grown in RPMI medium supplemented with 10% fetal calf serum followed by exposure to L. donovani promastigotes at a ratio of 50/1 (parasites/macrophages). The infection was performed for 24 hours at 37°C and with 5% CO₂. Non-internalized promastigotes were then removed by washing and infection continued for an additional 48 hours. Infected cells were then removed after addition of EDTA/trypsin (three minutes) followed by centrifugation at 1000g for 10 minutes. Pellets were then disrupted with electrophoresis loading buffer and used in Western blot analysis. The blots were blocked overnight at 4°C with tris-buffered saline with 0.1% Tween 20 (TBS-T) containing 1% bovine serum albumin (BSA) and subsequently probed with a pool of biotinylated monoclonal IgG antibodies specific for Ld-mad1. After several rinses with TBS-T, streptavidin labeled with horseradish peroxidase (Thermo Scientific Pierce, Rockford, IL) was added. After additional washings, bound conjugates were detected using the ECL enhanced chemiluminescence system (Amersham/GE Healthcare, Piscataway, NJ) and proteins were visualized on a ChimiDocTouch Imaging System from BioRad, Hercules, CA.

ELISA

A capture ELISA antigen detection test was developed using purified mAb specific for different epitopes in each leishmanial biomarker. Briefly, 96 well ELISA plates (EIA/RIA plate High Binding, Corning International, Corning, NY) were coated overnight at 4°C with mAbs in PBS at a concentration of 2ug/ml. The multiplexed ELISA, designed to detect simultaneously Li-isd1, Li-txn1, and Li-ntf2, Ld-mao1, Ld-ppi1, and Ld-mad1 was assembled using as capture reagent a pool of the six specific mAbs (mAbP11, mAbP212, mAbP33, mAbP41, mAbP64 and mAbP713) at 2ug/ml each. Wells were washed with PBS+ 0.1% Tween-20 (Sigma Chemical Co., St. Luis, MO) and blocked at room temperature with PBS + 1% BSA + 0.1% Tween 20 (PBS/BSA/Tween) for 2h. After washing, recombinant antigen or human urine samples were added and incubated overnight at 4°C. Plates were washed, followed by incubation for 1h with a biotin labeled mAb that recognizes a different epitope than that recognized by the mAb that was used for capture. For the multiplex assay, the developing reagent consisted of a pool of the biotinylated mAbs used in the individual assays (mAbP115, mAbP218, mAbP311, mAbP49, mAbP618, and mAbP79). Following several rinses in PBS/BSA/Tween, peroxidase-labeled streptavidin at 1:2,000 dilution (BD Bioscience, Franklin Lakes, NJ) was added for 30 minutes. The plates were then washed and reactions were developed with TMB Microwell Peroxidase substrate system (KPL, Gaithersburg, MD 20878) and read at 450 nm.

Results

Leishmanial protein biomarkers present in urine of VL patients

In recent years we have discovered and described several L. infantum and L. donovani proteins excreted in the urine of patients with VL from Brazil, Kenya and India [22-25]. Six of them have been selected as targets for the development of a diagnostic antigen detection assay. The selection criteria for these peptides included: a high MS XCorr >3.0; BLAST analysis of the peptide matching the pathogen's biomarker having a highly significant E-value; finding two or more peptides per donor protein; found in urine samples from at least two out of ten patients; and ideally with a MW \leq 30–40 kDa. Table 1 describes these selected biomarkers, which were previously discovered in the urine of VL patients from Brazil, Kenya and India [22-25]. As can be seen, three of the six biomarkers were found in the urine of VL patients from these three countries and three were found only in urine of patients from Kenya and India. With the exception of *Ld-mad1*, we have tested the possible utility of these biomarkers for the initial development of a proof of concept urine-based antigen detection assay for the diagnosis of VL. We produced and purified pairs of polyclonal antibodies (rabbit IgG and chicken IgY) specific for each of these biomarkers and used them to assemble capture ELISA to detect them in the urine of VL patients and controls. A multiplexed assay assembled for the detection of *Li-isd1*, Li-txn1 and Li-ntf2 originally discovered in urine samples of VL patients from Brazil (where the disease is caused by L. infantum) proved to be highly sensitive and specific for the diagnosis of VL occurring in patients from this country (~95% and 100% respectively). In contrast, the test was less accurate for the diagnosis of VL occurring in patients from Kenya and India, where the disease is caused by L. donovani. However, the inclusion of Ld-mao1 and Ld-ppi1 (previously discovered in the urine of VL patients from Kenya and India) improved the sensitivity of the test to nearly clinically acceptable levels of 82% [22].

Unfortunately, at that time we were unable to assemble a multiplexed test that included *Ld-mad1*, which is a biomarker that was present in several urine samples from VL patients from

Leishmanial donor protein of discovered peptide (putative)	Accession	Peptide identified in patient urine	MW of donor protein	Frequency in urine samples of patients from:		
					Kenya	India
Iron superoxide dismutase (<i>Li-isd1</i>)	XM_001467829	K.LNAAAESNSGLASK.S R.GGGEPSGPLASAIVDSFGSFASFK.K	21.53	1/5	5/6	1/9
Tryparedoxin (<i>Li-txn1</i>)	XM_001466605	K.HLGDVLK.L K.QNDMVDMSSLSGK.T K.LVEFYEK.H K.MPWLSIPFEK.R R.NVVEALTK.Q K.QYKVESIPTLIGLNADTGDTVTTR. A R.HALTQDPEGEQFPWRDE	16.7	2/5	5/6	5/9
Nuclear transport factor 2 (<i>Li-ntf2</i>)	XM_001463701	R.DQLAGIYRPNTLLTWQK.E K.EQVQGVDAIMAR.F R.FANLGFTEAAFK.Q	13.89	2/5	4/6	4/9
Maoc family dehydratase (<i>Ld-mao1</i>)	XP_003858460.1	K.EMPGPGTVYLSQNLR.F K.GLISLSNIVQK.T K.TLQFEGESEWSVTK	16.97	Not found	2/6	2/9
Peptidyl-prolyl Isomerase (<i>Ld-ppi1</i>)	XP_003858557.1	R.RTGQPTTISYEEAVTELQK.W R.VTFEEAAR.Q	12.62	Not found	2/6	1/9
Malate dehydrogenase (Ld-mad1)	XP_003864180.1	R.VAVLGAAGGIGQPLSLLLK.N R.DDLFNTNASIVR.D R.LFGVTTLDVVR.A R.IQFGGDEVVK.A	33.28	Not found	3/6	6/9

Table 1. Leishmania peptides previously identified by mass spectroscopy in individual urine samples of patients with VL from Brazil, Kenya and India.

https://doi.org/10.1371/journal.pntd.0008246.t001



Fig 2. SDS-PAGE (A) and Western blot (B) analysis of purified recombinant *Ld-mad1.* Recombinant protein containing His-tag amino terminal residues were expressed in *E. coli* BL21(DE3)pLysS followed by purification by affinity chromatography using Ni-NTA agarose matrix. Purity was evaluated by SDS-PAGE (4–20% gradient polyacrylamide gel) Coomassie blue staining (A) and by Western blot using anti-His-tag mAb (B). Arrows point to a protein band with MW that matches the deduced MW of *Ld-mad1.* Several other bands of higher MW are seen in the Western blot; since these bands are revealed with an anti-His tag mAb they are possibly aggregates or polymers of *Ld-mad1.*

https://doi.org/10.1371/journal.pntd.0008246.g002

Kenya and India (Table 1). In order to include this biomarker in the multiplexed assay, we produced and purified the recombinant molecule to be used for the generation of specific antibodies. Before immunization, the purity of the recombinant *Ld-mad1* was assessed and confirmed by SDS-PAGE with Coomassie blue staining and Western blot analysis (Fig 2).

Western blot analysis using specific monoclonal antibodies (below) and whole cell extract from both *L. donovani* promastigotes and amastigotes confirmed that the purified *Ld-mad1* protein matches a genuine leishmanial molecule. Both stages of the parasite cell cycle were included because previous publications suggested that *Ld-mad1* is primarily expressed by amastigotes [33, 34]. Western blot analysis confirmed that the native *Ld-mad1* protein is barely detected in promastigote forms of the parasite and is clearly expressed by amastigotes (Fig.3). Importantly, these results validate that the recombinant *Ld-mad1* protein is an accurate copy of the native molecule produced by *L. donovani*.

Production and characterization of monoclonal antibodies specific for the leishmanial biomarkers *Li-isd1*, *Li-txn1*, *Li-ntf2*, *Ld-mao1*, *Ld-ppi1 and Ld-mad1*

Subsequent to immunization of mice and generation of hybridomas, we performed limiting dilution followed by determination of their reactivity with the six leishmanial biomarkers, which was tested by direct ELISA and Western blot. For each biomarker we selected 20 IgG positive hybridoma clones for mapping of epitope recognition. This step aimed to select clones



Fig 3. Immunochemical detection of *Ld-mad1* **in** *L. donovani* **amastigotes.** Validation of *Ld-mad1* as a genuine protein produced by *L. donovani* was performed by Western blot using amastigote and promastigote cell lysates from the parasite. *L. donovani*-infected macrophage DH82 was used as the source of the amastigote forms of the parasite. The antibody was a pool of three mAbs specific for *Ld-mad1*. *rLd-mad1*, purified recombinant protein; MWM, Markers and respective MWs; Lane 1, crude extract of macrophage DH82 infected with *L. donovani*; Lane 2, crude extract of non-infected macrophage DH82; Lane 3, crude extract of *L. donovani* promastigotes. The black arrow points to *Ld-mad1* recombinant protein band recognized by the mAbs. A protein band of the same MW as *Ld-mad1* that is present only in the cell lysate of *L. donovani* infected macrophages. No bands that correspond to the MW of *Ld-mad1* were seen in membranes blotted with the same amastigote and promastigote cell lysates and probed with an irrelevant mAb.

https://doi.org/10.1371/journal.pntd.0008246.g003

producing mAbs that would recognize different epitopes in each protein, which are essential for the assembling of a capture ELISA to detect the markers. The epitope mapping was performed by direct ELISA using synthetic 20mer peptides covering the entire full length of each biomarker and overlapping by 10 amino acids. This protocol allowed the detection of several mAbs (Table 2) that were specific for different epitopes spanning from the N to the C terminus of each biomarker. Several supernatants from the hybridoma clones did not clearly recognize the 20mer peptides. Fig 4 depicts the epitope recognition of the mAbs that recognized the 20mer peptides. The mAbs were designated by the prefix P1, P2, P3, P4, P6, P7 (for *Li-isd1, Li-txn1, Li-ntf2, Ld-mao1, Ld-ppi1 and Ld-mad1* respectively) followed by a number that corresponds to the hybridoma clones labeled from 1–20. The hybridoma clones that are listed in Fig 4 were then expanded for production and purification of the IgG mAbs. Approximately 10–15 mg of purified IgG were obtained for each clone. From each purified mAb, an aliquot of 2 mg was separated and biotinylated.

Biomarker	# of epitopes recognized by different mAbs
Li-isd1	7
Li-txn1	3
Li-ntf2	4
Ld-mao1	5
Ld-ppi1	4
Ld-mad1	6

Table 2. Number of generated hybridoma clones producing mAbs specific for different epitopes in each leishmanial biomarker.

https://doi.org/10.1371/journal.pntd.0008246.t002

To select the mAb producing hybridoma clones for the development of the final capture ELISA, each purified IgG mAb was tested either as a capture or developing reagent using a checkerboard approach, i.e., each of all mAbs specific for one epitope of one of the biomarkers was tested as a capture reagent, paired with all other purified biotinylated mAbs specific for the other epitopes of the same biomarker. Fig 5 illustrates the pairs of antibodies that provided the best signal to noise results for each biomarker. The criterion used for the selection of the best pairs of antibodies was primarily those combinations that would yield a sensitivity or limiting of dilution that was equal to or smaller than 45 pg/ml in a capture ELISA performed with urine spiked with the biomarker. Interestingly and surprisingly, the best combination of mAb



Fig 4. Mapping of epitope recognition by IgG mAbs specific to the leishmanial biomarkers *Li-isd1*, *Li-txn1*, *Li-ntf2*, *Ld-mao1*, *Ld-ppi1 and Ld-mad1*. IgG mAbs were purified from supernatants of cloned hybridoma cells that recognized different peptides on the leishmanial biomarkers and were subsequently biotin labeled. Mapping of epitope recognition was performed by direct ELISA using plates coated with synthetic purified 20mer peptides covering the entire full length of each biomarker and overlapping by 10 amino acids. Letters in red and blue represent a sequence of 10–15 amino acids or the epitopes that were recognized by the indicated mAbs. Blue brackets illustrate the pair of antibodies that were selected as the capture and detecting reagents of the assembled capture ELISAs.

https://doi.org/10.1371/journal.pntd.0008246.g004



Fig 5. Determination of the limit of detection (LOD) of capture ELISAs assembled to individually detect the protein biomarkers *Li-isd1*, *Li-txn1*, *Li-ntf2*, *Ld-mao1*, *Ld-ppi1 and Ld-mad1* spiked in urine samples of normal healthy subjects. Capture mAbs at previously determined concentration of 200 ng/well were used to coat the ELISA plates that were prepare to detect each biomarker individually. Wells were then incubated with various concentrations of the antigen diluted in urine from normal healthy subjects followed by incubation with biotin labeled developer mAb (2 µg/ml). Reactions were developed after addition of streptavidin labeled peroxidase, substrate (H₂O₂) and the chromophore TMB. Results are expressed as OD read at 450nm. Note that for all six assays the LOD was \leq 45 pg/ml of biomarker.

https://doi.org/10.1371/journal.pntd.0008246.g005

pairs comprised of antibodies that recognize epitopes that are almost contiguous to each other, in opposition to epitopes that are located far from each other in the protein molecule (Fig 4, areas highlighted with blue brackets).

Initial determination of clinical sensitivity of each individual biomarker assay

Given that the ultimate goal of this work is to develop a diagnostic test for both New and Old World VL, we proceeded with a preliminary clinical validation of the six leishmanial biomarker detection assays using urine from patients with VL from Brazil (n = 24) and Kenya (n = 45). Urine was collected prior to initiating VL therapy. As controls, 25 urine samples were collected from healthy individuals from VL-endemic countries and 10 urine samples from healthy individuals from Massachusetts, USA. All urine samples were frozen (-80° C) prior to testing and handled identically upon thawing. The cutoff to discriminate positive from negative results was calculated as the average of the results for the control samples plus 3 SD. The



Leishmanial biomarker

Fig 6. Capture ELISA for detection of *Li-isd1*, *Li-txn1*, *Li-ntf2*, *Ld-mao1*, *Ld-ppi1* and *Ld-mad1* in urine of VL patients and local healthy controls from Brazil. ELISA was assembled as indicated in Fig 5. Urine samples were VL patients (n = 24) and healthy control subjects (n = 10). (A) Results represent the sensitivity of the assays performed to detect each individual biomarker. (B) Is a representation of the combined best results obtained for each of the urine samples that was positive (or not) with at least one of the individual biomarker assays. Red solid lines represent the cutoff values, which were calculated using the average of the OD obtained from the urine of normal healthy control subjects + 3 SD. These are representative results of at least three experiments performed at different times with the same urine samples and same capture ELISA.

https://doi.org/10.1371/journal.pntd.0008246.g006

standardization of the individual capture ELISA for each of the six markers was performed "blindly", i.e., the test for each individual marker using each individual urine sample was performed without the knowledge of the results obtained for the other markers.

The results showed that the sensitivities for the assays for *Li-isd1*, *Li-txn1*, *Li-ntf2*, *Ld-mao1*, *Ld-ppi1 and Ld-mad1* using urine samples from VL patients from Brazil were 45%, 33.3%, 29.1%, 41.6, 29.1% and 0% respectively (Fig 6).

The sensitivities of the individual assays performed with VL patients from Kenya were 20%, 35.5%, 31%, 44.4%, 40% and 20% respectively (Fig 7).

Importantly, an in-depth analysis of the results of each individual assay revealed that 0–5 biomarkers could be detected in each patient's urine sample from either Brazil or Kenya (Tables 3 and 4 respectively) and that the combined sensitivity of these markers is 91.66% and 93.33% for VL from these two countries respectively. At this time point, we did not attempt to correlate the results with disease severity because this evaluation is beyond the scope of the present study and would require a much larger patient sample size and a specific Human Subject Protocol approval, which would be different from the ones approved for the current study. Nonetheless, the results highlight and confirm our previous observation that a highly sensitive assay would need to include reagents that detect several different biomarkers preferentially assembled in a multiplexed format.



Leishmanial biomarker

Fig 7. Capture ELISA for detection of *Li-isd1*, *Li-txn1*, *Li-ntf2*, *Ld-mao1*, *Ld-ppi1 and Ld-mad1* in urine of VL patients and local healthy controls from Kenya. ELISA was assembled as indicated in Fig 5. Urine samples were from VL patients (n = 45) and healthy control subjects (n = 24). (A) Results represent the sensitivity of the assays performed to detect each individual biomarker. (B) Is a representation of the combined best results obtained for each of the urine samples that was positive (or not) with at least one of the individual biomarker assays. Solid red lines represent the cutoff values, which were calculated using the average of the OD obtained from the urine of normal healthy control subjects + 3 SD. These are representative results of at least three experiments performed at different times with the same urine samples and same capture ELISA.

https://doi.org/10.1371/journal.pntd.0008246.g007

Configuration and initial clinical validation of a multiplexed assay

A multiplexed assay offers several advantages over single assays; these include the ability to detect several biomarkers per test (which is crucial to achieving a high degree of sensitivity), the speed of performing the test and lower cost to produce and use compared to detecting individual biomarkers.

The multiplexed ELISA designed to simultaneously detect *Li-isd1*, *Li-txn1*, *Li-ntf2*, *Ld-mao1*, *Ld-ppi1* and *Ld-mad1* was assembled using a pool of the six specific mAbs that were used in the individual assays (mAbP11, mAbP212, mAbP33, mAbP41, mAbP64 and mAbP713) as capture reagents. Similarly, the developing reagent consisted of a pool of the bio-tinylated mAbs used in the individual assays (mAbP115, mAbP218, mAbP311, mAbP49, mAbP618 and mAbP79). A preliminary evaluation of the multiplexed assay revealed no loss of sensitivity when compared to the individual assays (S1 Fig). To begin the clinical validation of the multiplexed assay we used the same urine samples from VL patients from Brazil and Kenya that were used to validate the individual capture ELISAs. In addition, several control

Table 3. Frequency of individual leishmanial biomarkers detected by capture ELISA in 24 urine samples obtained from VL patients from Brazil.

	# of detected markers per urine sample								
	None	1	2	3	4	5	6		
# of urine samples	2	8	8	5	1	None	None		

https://doi.org/10.1371/journal.pntd.0008246.t003

		# of detected markers per urine sample								
	None	1	2	3	4	5	6			
# of urine samples	3	17	11	9	4	1	None			

Table 4. Freque	ıcy of individual le	ishmanial biomarke	ers detected by	v capture ELISA in	45 urine sampl	es obtained
from VL patients	s from Kenya.					

https://doi.org/10.1371/journal.pntd.0008246.t004

samples from non-VL patients who had other infectious diseases (cutaneous leishmaniasis, n = 6; Chagas disease, n = 6; schistosomiasis, n = 6; and tuberculosis, n = 12) were also tested. The results are illustrated in Fig.8 and show that the multiplexed assay has excellent sensitivity for the diagnosis of VL from Brazil (91.66%) and Kenya (93.33%). Interestingly, these sensitivities were the same as those observed for the calculated combined individual assays with the same samples giving positive/negative results. Importantly, the multiplexed assay had a



Disease condition

Fig 8. Initial clinical validation of a multiplexed assay for the diagnosis of VL from Brazil and from Kenya. ELISA plates were coated with a pool of affinity purified antibodies specific for all six biomarkers (*Li-isd1*, *Li-txn1*, *Li-ntf2*, *Ld-mao1*, *Ld-ppi1 and Ld-mad1*) followed by blocking and overnight incubation with urine samples from VL patients from Brazil (n = 24), VL patients from Kenya (n = 45) and urine samples from non-VL patients with the following diseases: CL (cutaneous leishmaniasis), n = 6; CD (Chagas disease), n = 6; Sch (schistosomiasis), n = 6; and TB (tuberculosis), n = 12. In addition, urine from 35 healthy control subjects was also included. Plates were washed and wells were incubated with streptavidin labeled peroxidase, the substrate H_2O_2 and the chromophore TMB. OD was then read at 450nm. The dashed red line represents the cutoff value (0.26) calculated as described in the legend of Fig 6. This is a representative result of at least three experiments performed at different times with the same urine samples.

https://doi.org/10.1371/journal.pntd.0008246.g008

specificity of 100% since no positive result was observed with urine samples from healthy control subjects or from patients having other infectious diseases. Therefore, the multiplexed assay assembled with mAbs specific for *Li-isd1*, *Li-txn1*, *Li-ntf2*, *Ld-mao1*, *Ld-ppi1* and *Ld-mad1* confirms and expands our former proof-of-concept observations made with polyclonal antibodies.

Discussion

Visceral leishmaniasis (VL) or kala-azar is a systemic parasitic disease that is endemic in 75 countries with more than 500 million people at risk of infection. It is estimated to affect 50,000–90,000 people each year, with 90% of the cases occurring in India, Kenya, Somalia, Sudan, South Sudan and Brazil [2]. VL is a fatal disease if not treated [1, 2]. Laboratory diagnosis of this disease can be performed using several different approaches. However, there is still a need for the development of a simple, non-expensive, sensitive and specific test that can be used worldwide not only for the accurate diagnosis of this serious disease but also to monitor the treatment efficacy [35].

In recent years, we have shown that protein antigens from the pathogens *L. infantum* and *L. donovani* could be detected in urine of patients with active VL using mass spectroscopy. The discovery of these pathogen proteins led us to the development of a urine-based capture ELISA for the diagnosis of this serious disease. This ELISA, which was assembled with polyclonal antibodies raised in rabbits and chickens, constitutes a successful proof of concept for the development of a urine-based diagnostic test for VL. A multiplexed assay based on five of the biomarkers discovered could be developed and shown to be highly sensitive and specific for the diagnosis of VL [23–25]. However, the use of polyclonal antibodies is a barrier to scaling-up such a test for commercial production. A proven alternative is the use of mAbs, which circumvents this limitation. Here, we report our results on the production and use of mAbs specific for the *L. donovani / L. infantum* antigens *Li-isd1, Li-txn1, Li-ntf2, Ld-mao1, Ld-ppi1 and Ld-mad1* for the development of a sensitive and specific multiplexed test for the diagnosis of VL.

Our initial efforts were concentrated on the validation of the biomarker Ld-mad1 (putative L. donovani malate dehydrogenase) because producing both purified recombinant protein and specific rabbit and chicken antibodies to detect this molecule proved difficult in our former study. In the present study, we opted to produce monoclonal antibodies (mAbs) to Ld-mad1 to circumvent the non-specific reactivity observed with the polyclonal antibodies generated against this molecule. The generation of the mAbs using a conventional procedure was readily achieved and 20 clones producing antibodies that were highly specific for Ld-mad1 were selected using Western blot analysis. Specificity was determined by the detection of a strong band that matched the expected MW of Ld-mad1. This was easily obtained regardless of the purity of the recombinant protein used in the Western blot and to immunize the mice. Furthermore, the reactivity and specificity of the mAbs to the native Ld-mad1 protein (in contrast to the recombinant molecule) was also determined by Western blot analysis using leishmania parasite cell extract. The Leishmania malate dehydrogenase protein was originally described as preferentially being produced by amastigote forms of Leishmania mexicana [33, 34]. Therefore, to validate the reactivity of the mAbs with native Ld-mad1 we performed Western blot analysis using both amastigote and promastigote forms of L. donovani. The results clearly confirmed that L. donovani malate dehydrogenase (Ld-mad1), similarly to its production by L. mexicana mexicana, is preferentially or only expressed by L. donovani amastigotes.

The generation, selection, and purification of mAbs to all other five biomarkers were also obtained without difficulty. For practical reasons, only IgG producing clones were selected. Between 4–7 clones per biomarker were obtained. For each marker we were able to purify 15–

20 mg of specific IgG from 3–7 clones, which were used for the selection of pairs of antibodies for assembling the capture ELISA to detect each biomarker. Interestingly and unexpectedly, the best antibody pairs for each of all six biomarkers recognized linear epitopes that were almost contiguous to each other in opposition to distantly located epitopes in the amino acid sequence of each molecule. At this point, we do not have an explanation for this observation, but it is possible that the binding of an antibody to a given linear epitope will partially change the tertiary structure of the protein, which might result in the exposure of contiguous but not distant linear epitopes of the molecule. Although an interesting observation, we did not explore it any further because it is beyond the scope of this study.

The ultimate goal of this work was to translate our preliminary proof of concept observation, i.e., the development of an antigen detection diagnostic test for VL using polyclonal rabbit IgG and chicken IgY, into a test that instead uses immortalized mAbs specific for each of six leishmanial biomarkers that we have previously described. This former work, demonstrated the proof of concept that an antigen detection test could be developed for the diagnosis of VL. Unfortunately, because that study was performed with polyclonal antibodies the results could not be directly translated into a de facto clinical test for VL. In contrast, the current work presents data that supports the immediate translation of the results into a new clinically useful diagnostic test for VL. Immortalized mAbs, in contrast to polyclonal antibodies, are suitable reagents for mass scale-up and continuous production of the final test. Initial validation of the ELISAs assembled to detect each one of the six biomarkers confirmed that the selected mAb pairs had an excellent biochemical sensitivity 15-45 pg/ml. Importantly, urine causes little or no interference in the assay sensitivity when compared to assays performed with the biomarkers diluted in PBS/BSA (S2 Fig). The clinical sensitivity of each assay confirmed our previous observations, that no single marker is sufficient for the development of a sensitive assay. Indeed, with the exception of the assay to detect Ld-mad1, the individual sensitivity varied from 20% to 50% of the VL patients from either Brazil or Kenya. In contrast, the assay to detect Ld-mad1 was negative with all urine samples from VL from Brazil but had a sensitivity of 20% among the urine samples from VL patients from Kenya. It is possible that this lower sensitivity to detect Ld-mad1 is due to the fact that this molecule has a higher MW than the other biomarkers, which may limit its excretion in the urine as a full-length protein, thus complicating its detection by the capture ELISA. Also intriguing was that fact that Ldmad1 was not detected in any of the 24 urine samples from VL patients from Brazil. At this point we do not have a clear explanation for this results. However, it is possible that it could be due to the limited number of samples that was thus far used for this validation. It is also theoretically possible that *Ld-mad1* is not eliminated in the urine of patients of VL caused by *L*. infantum.

Nonetheless, because many urine samples were positive for several of the six markers (Tables $\underline{3}$ and $\underline{4}$) it resulted in a combined sensitivity of 91.66% and 93.33% for the Brazil and Kenya urine samples respectively. These results concur with our previous observations indicating that a highly sensitive test requires the simultaneous detection of several VL biomarkers, either separately or in a multiplex format [23].

Indeed, the mAb-based multiplexed assay assembled to detect all six markers simultaneously showed excellent sensitivity and specificity. The overall sensitivity was again 91.66% for VL from Brazil and 93.33% for VL from Kenya, which confirms the calculated combined sensitivity of the individual assays and validates the multiplexed assay. Importantly, the sensitivity of this urine-based test is much higher than that observed for the invasive tests that detect the parasites themselves (~50%) and similar to that of methods that detect parasite nucleic acids, which is approximately 90% [4, 11, 36]. Moreover, the overall specificity of the multiplexed test was 100% when using urine samples not only from healthy control subjects but from non-VL patients who have other infectious diseases (cutaneous leishmaniasis, Chagas disease, schistosomiasis, and tuberculosis). We recognize that the sensitivity/specificity of the assembled multiplexed assay thus far is based on a limited number of urine samples from both patients and controls. Nonetheless, these results are very encouraging. We are in the process of expanding the clinical validation of this promising new test for the diagnosis of VL in order to translate it into a clinical tool. This validation will include a much larger sample size and will have patients not only from Brazil and Kenya but from other endemic countries as well, such as India and Bangladesh. In addition, we are evaluating the suitability of the mAbs developed for assembly into a reliable immunochromatographic rapid test for point-of-care diagnosis of VL.

Finally, because the multiplexed assay described here is assembled with highly defined and specific mAbs, this test by definition, is much more accurate and specific than assays assembled with polyclonal antibodies. Therefore, the new mAb-based multiplexed assay described in this communication will be of great utility not only for the diagnosis of active VL but also as an important tool to monitor the treatment efficacy for this disease. In fact, we have preliminary proof-of-concept evidence for this possible latter use from an ELISA assembled with conventional rabbit and chicken antibodies [25]. Moreover, the new mAb-based multiplexed assay should be a useful resource for diagnosis of other clinical forms and/or the severity of the disease e.g., asymptomatic VL, post kala-azar dermal leishmaniasis, VL/HIV co-infection, etc.

Supporting information

S1 Fig. Sensitivity of a multiplexed capture ELISA assembled to detect six Leishmania biomarkers (antigens) compared to capture ELISAs assembled to individually detect each bio**marker.** For single antigen detection capture ELISA (A), plates were coated with a single mAb antibody specific for one of the following leishmanial biomarkers: iron superoxide dismutase 1 (*Li-isd1*), tryparedoxin 1 (*Li- trx1*), nuclear transport factor 2 (*Li-ntf2*), maoC dehydralase (*Ld*mao1), peptidyl-prolyl cis-trans isomerase (Ld-ppi1) and malate dehydrogenase (Ld-mad1) an individual biomarker. For multiplexed capture ELISA (B) plates were coated with a pool containing mAbs specific for the six biomarkers. For the single assay, detection was performed using a single biotinylated mAb specific for a different epitope than that recognized by the mAb used to coat the plates. For the multiplexed assay, detection was performed using a pool containing biotinylated mAbs that were specific for different epitopes of each individual antigen that were recognized by the pool of mAbs used to coat the plates. Reactions were developed after addition of peroxidase streptavidin A plus the substrate H_2O_2 and the chromophore TMB. Results are expressed as OD read at 450nm. Note that there is no loss of sensitivity of the multiplexed assay compared to the single assay. (PDF)

S2 Fig. Sensitivity of a multiplexed capture ELISA for detection of leishmanial biomarkers (antigens), spiked in either buffer or in urine of healthy subjects. A pool, containing mAbs specific for the markers *Li-isd1*, *Li-txn1*, *Li-ntf2*, *Ld-mao1*, *Ld-ppi1*, and *Ld-mad1*, each at 2 μ g/ml was used to coat the ELISA plates. Wells were then incubated with various concentrations of the six makers diluted either in PBS plus 1% BSA or in urine from healthy subjects. Detection was performed using a pool of biotinylated mAbs that were specific for different epitopes of each individual marker that were recognized by the pool of mAbs used to coat the plates. Reactions were developed after addition of peroxidase streptavidin A plus the substrate H₂O₂ and the chromophore TMB. Results are expressed as OD read at 450nm. Note that urine does not interfere with the sensitivity of the multiplexed assay. (PDF)

Acknowledgments

For its overall mission, DND*i* received financial support from UK aid, UK; Médecins sans Frontières International and the Swiss Agency for Development and Cooperation (SDC), Switzerland. A full list of DND*i*'s donors can be found at http://www.dndi.org/donors/donors

Author Contributions

Conceptualization: Claudia Abeijon, Antonio Campos-Neto.

Formal analysis: Claudia Abeijon, Fabiana Alves, Séverine Monnerat, Ricardo T. Fujiwara, Antonio Campos-Neto.

Funding acquisition: Claudia Abeijon, Antonio Campos-Neto.

- **Investigation:** Claudia Abeijon, Fabiana Alves, Séverine Monnerat, Agostinho G. Viana, Raquel M. Almeida, Lilian L. Bueno, Ricardo T. Fujiwara, Antonio Campos-Neto.
- Methodology: Claudia Abeijon, Agostinho G. Viana, Raquel M. Almeida, Antonio Campos-Neto.
- Project administration: Claudia Abeijon, Antonio Campos-Neto.
- **Resources:** Claudia Abeijon, Fabiana Alves, Séverine Monnerat, Jane Mbui, Agostinho G. Viana, Lilian L. Bueno, Ricardo T. Fujiwara, Antonio Campos-Neto.
- Supervision: Fabiana Alves, Jane Mbui, Lilian L. Bueno, Ricardo T. Fujiwara, Antonio Campos-Neto.
- Validation: Claudia Abeijon, Fabiana Alves, Antonio Campos-Neto.
- Writing original draft: Claudia Abeijon, Antonio Campos-Neto.
- Writing review & editing: Claudia Abeijon, Fabiana Alves, Séverine Monnerat, Antonio Campos-Neto.

References

- Alvar J, Velez ID, Bern C, Herrero M, Desjeux P, Cano J, et al. Leishmaniasis worldwide and global estimates of its incidence. PLoS ONE. 2012; 7(5):e35671. https://doi.org/10.1371/journal.pone.0035671 PMID: 22693548
- 2. WHO. Weekly Epidemiological Record. 2018; 93:521-40.
- Dixit KK, Verma S, Singh OP, Singh D, Singh AP, Gupta R, et al. Validation of SYBR green I based closed tube loop mediated isothermal amplification (LAMP) assay and simplified direct-blood-lysis (DBL)-LAMP assay for diagnosis of visceral leishmaniasis (VL). PLoS neglected tropical diseases. 2018; 12(11):e0006922. https://doi.org/10.1371/journal.pntd.0006922 PMID: 30439953
- 4. Antinori S, Calattini S, Longhi E, Bestetti G, Piolini R, Magni C, et al. Clinical use of polymerase chain reaction performed on peripheral blood and bone marrow samples for the diagnosis and monitoring of visceral leishmaniasis in HIV-infected and HIV-uninfected patients: a single-center, 8-year experience in Italy and review of the literature. Clin Infect Dis. 2007; 44(12):1602–10. <u>https://doi.org/10.1086/518167</u> PMID: 17516404
- Costa JM, Durand R, Deniau M, Rivollet D, Izri M, Houin R, et al. PCR enzyme-linked immunosorbent assay for diagnosis of leishmaniasis in human immunodeficiency virus-infected patients. J Clin Microbiol. 1996; 34(7):1831–3. PMID: 8784604
- Cota GF, de Sousa MR, Rabello A. Predictors of visceral leishmaniasis relapse in HIV-infected patients: a systematic review. PLoS Negl Trop Dis. 2011; 5(6):e1153. <u>https://doi.org/10.1371/journal.pntd.</u> 0001153 PMID: 21666786
- Guerin PJ, Olliaro P, Sundar S, Boelaert M, Croft SL, Desjeux P, et al. Visceral leishmaniasis: current status of control, diagnosis, and treatment, and a proposed research and development agenda. Lancet Infect Dis. 2002; 2(8):494–501. https://doi.org/10.1016/s1473-3099(02)00347-x PMID: 12150849

- Melby PC. Recent developments in leishmaniasis. Curr Opin Infect Dis. 2002; 15(5):485–90. <u>https://doi.org/10.1097/00001432-200210000-00005 PMID: 12686880</u>
- Riera C, Fisa R, Ribera E, Carrio J, Falco V, Gallego M, et al. Value of culture and nested polymerase chain reaction of blood in the prediction of relapses in patients co-infected with leishmania and human immunodeficiency virus. Am J Trop Med Hyg. 2005; 73(6):1012–5. PMID: 16354803
- Srivastava P, Dayama A, Mehrotra S, Sundar S. Diagnosis of visceral leishmaniasis. Trans R Soc Trop Med Hyg. 2011; 105(1):1–6. https://doi.org/10.1016/j.trstmh.2010.09.006 PMID: 21074233
- Srivastava P, Mehrotra S, Tiwary P, Chakravarty J, Sundar S. Diagnosis of Indian visceral leishmaniasis by nucleic acid detection using PCR. PLoS ONE. 2011; 6:e19304. https://doi.org/10.1371/journal. pone.0019304 PMID: 21559398
- Sundar S, Rai M. Laboratory diagnosis of visceral leishmaniasis. Clin Diagn Lab Immunol. 2002; 9 (5):951–8. https://doi.org/10.1128/CDLI.9.5.951-958.2002 PMID: 12204943
- Braz RF, Nascimento ET, Martins DR, Wilson ME, Pearson RD, Reed SG, et al. The sensitivity and specificity of Leishmania chagasi recombinant K39 antigen in the diagnosis of American visceral leishmaniasis and in differentiating active from subclinical infection. Am J Trop Med Hyg. 2002; 67(4):344–8. https://doi.org/10.4269/ajtmh.2002.67.344 PMID: 12452487
- Badaro R, Benson D, Eulalio MC, Freire M, Cunha S, Netto EM, et al. rK39: a cloned antigen of Leishmania chagasi that predicts active visceral leishmaniasis. J Infect Dis. 1996; 173(3):758–61. <u>https://doi.org/10.1093/infdis/173.3.758</u> PMID: 8627048
- Badaro R, Eulalio MC, Benson D, Freire M, Miranda JC, Pedral-Sampaio D, et al. Sensitivity and specificity of a recombinant Leishmania chagasi antigen in the serodiagnosis of visceral leishmaniasis. Arch Inst Pasteur Tunis. 1993; 70(3–4):331–2. PMID: 7802487
- Bangert M, Flores-Chavez MD, Llanes-Acevedo IP, Arcones C, Chicharro C, Garcia E, et al. Validation of rK39 immunochromatographic test and direct agglutination test for the diagnosis of Mediterranean visceral leishmaniasis in Spain. PLoS neglected tropical diseases. 2018; 12(3):e0006277. <u>https://doi.org/10.1371/journal.pntd.0006277</u> PMID: 29494596
- Boelaert M, El-Safi S, Hailu A, Mukhtar M, Rijal S, Sundar S, et al. Diagnostic tests for kala-azar: a multi-centre study of the freeze-dried DAT, rK39 strip test and KAtex in East Africa and the Indian subcontinent. Trans R Soc Trop Med Hyg. 2008; 102(1):32–40. <u>https://doi.org/10.1016/j.trstmh.2007.09</u>. 003 PMID: 17942129
- Attar ZJ, Chance ML, El-Safi S, Carney J, Azazy A, El-Hadi M, et al. Latex agglutination test for the detection of urinary antigens in visceral leishmaniasis. Acta Trop. 2001; 78(1):11–6. https://doi.org/10. 1016/s0001-706x(00)00155-8 PMID: 11164745
- Chappuis F, Rijal S, Jha UK, Desjeux P, Karki BM, Koirala S, et al. Field validity, reproducibility and feasibility of diagnostic tests for visceral leishmaniasis in rural Nepal. Trop Med Int Health. 2006; 11(1):31– 40. https://doi.org/10.1111/j.1365-3156.2005.01533.x PMID: 16398753
- Sarkari B, Chance M, Hommel M. Antigenuria in visceral leishmaniasis: detection and partial characterisation of a carbohydrate antigen. Acta Trop. 2002; 82(3):339–48. <u>https://doi.org/10.1016/s0001-706x</u> (02)00043-8 PMID: 12039673
- Sundar S, Agrawal S, Pai K, Chance M, Hommel M. Detection of leishmanial antigen in the urine of patients with visceral leishmaniasis by a latex agglutination test. Am J Trop Med Hyg. 2005; 73(2):269– 71. PMID: 16103587
- 22. Abeijon C, Alves F, Monnerat S, Wasunna M, Mbui J, Viana AG, et al. Development of a Multiplexed Assay for Detection of Leishmania donovani and Leishmania infantum Protein Biomarkers in Urine Samples of Patients with Visceral Leishmaniasis. J Clin Microbiol. 2019; 57(5).
- Abeijon C, Campos-Neto A. Potential non-invasive urine-based antigen (protein) detection assay to diagnose active visceral leishmaniasis. PLoS Negl Trop Dis. 2013; 7(5):e2161. <u>https://doi.org/10.1371/journal.pntd.0002161</u> PMID: 23738023
- Abeijon C, Kashino SS, Silva FO, Costa DL, Fujiwara RT, Costa CH, et al. Identification and diagnostic utility of Leishmania infantum proteins found in urine samples from patients with visceral leishmaniasis. Clin Vaccine Immunol. 2012; 19(6):935–43. https://doi.org/10.1128/CVI.00125-12 PMID: 22518013
- Abeijon C, Singh OP, Chakravarty J, Sundar S, Campos-Neto A. Novel antigen detection assay to monitor therapeutic efficacy of visceral leishmaniasis. Am J Trop Med Hyg. 2016; 95(4):800–2. https://doi. org/10.4269/ajtmh.16-0291 PMID: 27481058
- 26. Kashino SS, Abeijon C, Qin L, Kanunfre KA, Kubrusly FS, Silva FO, et al. Identification of Leishmania infantum chagasi proteins in urine of patients with visceral leishmaniasis: a promising antigen discovery approach of vaccine candidates. Parasite Immunol. 2012; 34(7):360–71. <u>https://doi.org/10.1111/j.1365-3024.2012.01365.x</u> PMID: 22443237

- Epstein N, Epstein M. Immunochemical techniques. Part I: Hybridoma technology and monoclonal antibodies. Methods in enzymology. 1986; 121:1–947. PMID: 3523121
- Kohler G, Milstein C. Continuous cultures of fused cells secreting antibody of predefined specificity. Nature. 1975; 256(5517):495–7. https://doi.org/10.1038/256495a0 PMID: 1172191
- Eliasson M, Andersson R, Olsson A, Wigzell H, Uhlen M. Differential IgG-binding characteristics of staphylococcal protein A, streptococcal protein G, and a chimeric protein AG. J Immunol. 1989; 142 (2):575–81. PMID: 2521350
- Bayer EA, Wilchek M. The use of the avidin-biotin complex as a tool in molecular biology. Methods of biochemical analysis. 1980; 26:1–45. https://doi.org/10.1002/9780470110461.ch1 PMID: 7392958
- Campos-Neto A, Suffia I, Cavassani KA, Jen S, Greeson K, Ovendale P, et al. Cloning and characterization of a gene encoding an immunoglobulin-binding receptor on the cell surface of some members of the family Trypanosomatidae. Infect Immun. 2003; 71(9):5065–76. https://doi.org/10.1128/IAI.71.9. 5065-5076.2003 PMID: 12933849
- Maia C, Rolao N, Nunes M, Goncalves L, Campino L. Infectivity of five different types of macrophages by Leishmania infantum. Acta Trop. 2007; 103(2):150–5. <u>https://doi.org/10.1016/j.actatropica.2007.06.</u> 001 PMID: 17626776
- Mottram JC, Coombs GH. Leishmania mexicana: subcellular distribution of enzymes in amastigotes and promastigotes. Experimental parasitology. 1985; 59(3):265–74. <u>https://doi.org/10.1016/0014-4894</u> (85)90081-5 PMID: 3158538
- Mottram JC, Coombs GH. Leishmania mexicana: enzyme activities of amastigotes and promastigotes and their inhibition by antimonials and arsenicals. Experimental parasitology. 1985; 59(2):151–60. https://doi.org/10.1016/0014-4894(85)90067-0 PMID: 2982638
- Chappuis F, Sundar S, Hailu A, Ghalib H, Rijal S, Peeling RW, et al. Visceral leishmaniasis: what are the needs for diagnosis, treatment and control? Nat Rev Microbiol. 2007; 5(11):873–82. <u>https://doi.org/ 10.1038/nrmicro1748</u> PMID: 17938629
- 36. Piarroux R, Gambarelli F, Dumon H, Fontes M, Dunan S, Mary C, et al. Comparison of PCR with direct examination of bone marrow aspiration, myeloculture, and serology for diagnosis of visceral Leishmaniasis in immunocompromised patients. J Clin Microbiol. 1994; 32(3):746–9. PMID: 8195388